Author's response to reviews

Title: Sleep hygiene intervention for youth aged 10 to 18 years with problematic sleep: a before-after pilot study of sleep and weight

Authors:

Evan Tan (evan.tan@otago.ac.nz)
Dione Healey (dionehealey@psy.otago.ac.nz)
Andrew Gray (andrew.gray@otago.ac.nz)
Dr Barbara Galland (barbara.galland@otago.ac.nz)

Version: 2 Date: 19 June 2012

Author's response to reviews:

The trial registration is now reported.
The title has changed slightly